Author's response to reviews

Title: Evidence-based practice implementation: The impact of public versus private sector organization type on organizational support, provider attitudes, and adoption of evidence-based practice

Authors:

Gregory A. Aarons (gaarons@ucsd.edu)
David M. Sommerfeld (dsommerfeld@ucsd.edu)
Christine M. Walrath-Greene
(Christine.M.Walrath-Greene@macrointernational.com)

Version: 5 Date: 17 November 2009

Author's response to reviews: see over
November 17, 2009

Martin Eccles
Editor - Implementation Science

Re: Manuscript ID 6204242002194095

Dear Dr. Eccles:

Please accept our revised manuscript “Evidence-based practice implementation: The impact of public versus private sector organization type on organizational support, provider attitudes, and adoption of evidence-based practice” for consideration for publication in *Implementation Science*. We have made the requested editorial changes as outlined below. Please contact me if you have any questions or concerns about this manuscript.

Sincerely,

Gregory A. Aarons, Ph.D.
Associate Professor of Psychiatry
University of California, San Diego
Tel: 858-966-7703 x3550
Cell: 858-335-0700

1) The figs are .bmp files so couldn’t be copy-edited. They’re not in TR12 font, and the authors fail to put a zero to the left of their decimals throughout the figures (i.e. p<.01 should be p<0.01). Please correct this as in the manuscript.

   We have edited the figures and put zero in the notation. We are uploading three Powerpoint files, one for each figure.

The references are not in journal style. This will have to be corrected and so may as well be done now.

   We have edited the references to conform to journal style.

The Table legends look as if they contain data. If so, this should be either added to Table footnotes or incorporated into the manuscript text.

   We examined the Table legends and they do not contain data.